Author's response to reviews

Title: Establishing oral health promoting behaviours in children - parents' views on barriers, facilitators and professional support: a qualitative study

Authors:

Denise Duijster (D.Duijster@acta.nl)
Maddelon de Jong-Lenters (maddelon.dejong@tno.nl)
Gijsbert H.W. Verrips (erik.verrips@tno.nl)
Cor van Loveren (C.v.Loveren@acta.nl)

Version: 3
Date: 18 June 2015

Author's response to reviews:

Dear editor,

The following formatting changes have now been addressed in the revised submission:
Registration number of ethical approval,
Patient anonymity,
Storage of data in a public repository,
Description of the interview guide and the theoretical framework for data collection.

We hope that all formatting requirements have now been adequately addressed.

Kind regards (on behalf of all authors),

Denise Duijster